Author's response to reviews

Title: Acute kidney injury induced by thrombotic microangiopathy in a patient with haemophagocytic lymphohistiocytosis

Authors:

Myoung Nam Bae (mnnm0402@catholic.ac.kr)
Dae Hun Kwak (dh-k@live.com)
Se Jun Park (psj6936@naver.com)
Bum Soon Choi (sooncb@catholic.ac.kr)
Cheol Whee Park (cheowlwee@catholic.ac.kr)
Yeong Jin Choi (mdyjchoi@catholic.ac.kr)
Jong Wook Lee (jwlee@catholic.ac.kr)
Chul Woo Yang (yangch@catholic.ac.kr)
Yong-Soo Kim (kimcmc@catholic.ac.kr)
Byung Ha Chung (Chungbh@catholic.ac.kr)

Version: 4
Date: 23 January 2015

Author's response to reviews: see over
January 15\textsuperscript{th}, 2015

Dear Editor-in-Chief, BMC Nephrology:

I, along with my coauthors, would like to ask you to consider the attached manuscript entitled “\textbf{Acute kidney injury induced by thrombotic microangiopathy in a patient with haemophagocytic lymphohistiocytosis}” for publication in \textit{BMC Nephrology} as a Case Report.

The paper deals with a case of renal thrombotic microangiopathy (TMA) associated hemophagocytic lymphohistiocytosis (HLH). Renal TMA is a rare pathologic finding. There are only a few reports of TMA developing in patients with HLH. In our case, the patient was admitted with fever and oliguria. She was diagnosed with HLH and renal TMA. Renal TMA associated with HLH was successfully treated with cytotoxic therapy that included etopocide and cyclosporine. We believe that the findings presented here are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by another journal. The participant provided informed consent. All the authors have approved the manuscript and agree with submission to your esteemed journal. There are no conflicts of interest to declare.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,

Byung Ha Chung, MD, PhD

Department of Internal Medicine, Seoul St. Mary’s Hospital, The Catholic University of Korea. 222 Banpodae-ro, Seocho-gu, Seoul 137-040, Korea

Tel: +82.2-2258-6066, Fax: +82.2-536-0323, E-mail: Chungbh@catholic.ac.kr